Author's response to reviews

Title: Long-term outcomes after acute primary angle closure in a White Caucasian population

Authors:

Walter Andreatta (andreattawalter@hotmail.com)
Ibrahim Elaroud (i_melaroud@hotmail.com)
Peter Nightingale (Peter.Nightingale@uhb.nhs.uk)
Maged Nessim (m.nessim@bham.ac.uk)

Version: 2 Date: 18 January 2015

Author's response to reviews: see over
Dear Editor,

many thanks for your considering our paper for publication.

We made the following changes as you suggested:

(1) Requesting name of ethics committee.
Response: the name of the ethics committee was specified (line 115).

(2) Please put your Competing Interest after the Conclusion Section.
Response: competing interests were moved accordingly (line 239).

(3) Author's Contribution.
Response: contributions stated (lines 241-243).

(4) Acknowledgments.
Response: added (line 245).

(5) Requesting for consent statement.
Response: added (line 76).

(6) Requesting for Line Numbering.
Response: added.

Kind regards,
Walter Andreatta & co-authors